Author's response to reviews

Title: Production of icaADBC-encoded polysaccharide intercellular adhesin and therapeutic failure, in pediatric patients with staphylococcal device-related infections.

Authors:

Bernardo Diemond-Hernandez (brucediemond@yahoo.com.mx)
Fortino Solorzano-Santos (fortino.solorzano@imss.gob.mx)
Blanca Leanos-Miranda (blancalm@yahoo.com.mx)
Leoncio Peregrino-Bejarano (leonpb73@yahoo.com)
Guadalupe Miranda-Novales (guadalupe.mirandan@imss.gob.mx)

Version: 4 Date: 12 February 2010

Author's response to reviews: see over
February 17th, 2009.

BMC Infectious Diseases,

Dear Editors,

We are enclosing the revised manuscript “Production of icaADBC-encoded polysaccharide intercellular adhesin and therapeutic failure, in pediatric patients with staphylococcal device-related infections”, with the following modifications, according to the comments by the reviewers.

Reviewer's report
Version: 3 Date: 9 December 2009

1. Table 2: S. lugdunensis instead of S. lugdonensis
2. All page numbers of references were re-written to unify criteria.
3. All ica genes in the figures were written altogether (without separation).
4. The figures were edited and legends now match the obtained bands.

Thank you very much for your attention,

Guadalupe Miranda-Novales
Corresponding author